Author's response to reviews

Title: Reducing salt intake for prevention of cardiovascular disease in high-risk patients: study protocol for a cluster randomized trial of advanced health education intervention in northern Thailand (RESIP-CVD study)
ISRCTN39416277

Authors:

Myo N Aung (dr.myonyeinaung@gmail.com)
Motoyuki Yuasa (moyuasa@juntendo.ac.jp)
Saiyud Moolphate (ritbkk@bkk7.loxinfo.co.th)
Supalert Nedsuwan (supalert@gmail.com)
Hidehiro Yokokawa (yokokawa@pa3.so-net.ne.jp)
Tsutomu Kitajima (kitajima@ks.kyorin-u.ac.jp)
Kazuo Minematsu (kminema@juntendo.ac.jp)
Susumu Tanimura (tanimura@hyo-med.ac.jp)
Hiroshi Fukuda (hiro----@kt.rim.or.jp)
Yoshimune Hiratsuka (yoshi-h@tkf.att.ne.jp)
Koichi Ono (kono@juntendo.ac.jp)
Sachio Kawai (chyabo@juntendo.ac.jp)
Eiji Marui (marui@juntendo.ac.jp)

Version: 3 Date: 31 July 2012

Author's response to reviews: see over
Cover letter

Reducing salt intake for prevention of cardiovascular disease in high-risk patients: study protocol for a cluster randomized trial of advanced health education intervention in northern Thailand
(RESIP-CVD study) ISRCTN39416277

Editorial email

Thank you for submitting your revised manuscript to Trials. I am delighted to inform you that the Associate Editor recommends that we accept your manuscript for publication in Trials. However, before we can formally accept the manuscript can you please ensure that the following formatting corrections are made:

1. Title: please ensure the title conforms to journal style for study protocol articles. The title should follow the format "Reducing...: study protocol for a cluster randomized trial in northern Thailand".

2. Further consideration of your manuscript is conditional on improvement of the English used. Please ensure particular attention is paid to the abstract.

Authors’ reply

In reference to the suggestion of Editors-in-Chief, the protocol was finally polished to conform to the style of Trials.

1. Title has been changed conforming to the style of study protocol articles.

Reducing salt intake for prevention of cardiovascular disease in high-risk patients: study protocol for a cluster randomized trial of advanced health education intervention in northern Thailand

2. Together with the revise manuscript, we presented the English Language edited file showing how we polished the English language of the manuscript with the help of Mr. Richard Lawrence Mann, BA European Studies, Bradford University, UK. Currently English Teacher, Lampang Rajabhat University, Thailand.

Moreover, the tables and figures were carefully checked and conformed to the Trial journal style. The protocol was checked to meet the outline in author’s guideline. All the sections and order of the sections were checked and rearranged according to the editorial suggestion.

Thank you very much.

Dr Myo Nyein Aung
Corresponding author